Faculty of Applied Science and Engineering, University of Toronto CSC181: Introduction to Computer Programming

Assignment 5: Connect Four, Part II

I/We declare that this assignment is my/our own work, and is in accordance with the University of Toronto Code of Behaviour on Academic Matters and the Code of Student Conduct.

Student #1 Student #2

Family name: ______________ ______________

Family name: ______________ ______________ Given name: ______________ ______________

Given name: ______________ ______________ Student #: ______________ ______________

Student #: ______________ ______________ Signature: ______________ ______________

Signature: ______________ ______________